**Author's response to reviews**

**Title:** Gene network analyses point to the importance of human tissue kallikreins in melanoma progression

**Authors:**

Waleska K Martins (wkerlenmartins@gmail.com)
Gustavo H Esteves (gesteves@uepb.edu.br)
Otávio M Almeida (o.almeida@clinicanomina.com.br)
Gisele G Rezze (ggrezze@hotmail.com)
Gilles Landman (gilles.landman@gmail.com)
Sarah M Marques (sarah.marques@hsl.org.br)
Alex F Carvalho (alex.carvalho@hcancer.org.br)
Luiz F L Reis (luiz.reis@hsl.org.br)
João P Duprat (jduprat@uol.com.br)
Beatriz S Stolf (biastolf@yahoo.com.br)

**Version:** 12 **Date:** 17 October 2011

**Author's response to reviews:** see over
MS: 1503872799573458

Gene network analyses point to the importance of human tissue kallikreins in melanoma progression

Waleska K Martins, Gustavo H Esteves, Otávio M Almeida, Gisele G Rezze, Gilles Landman, Sarah M Marques, Alex F Carvalho, Luiz Fernando L Reis, João P Duprat and Beatriz S Stolf

Dear Production Team,

Thank you for your fast reply concerning the final corrections of the paper. We hope to have addressed the requested major and minor changes in the manuscript uploaded in the journal page.

Best regards,

Waleska K Martins